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TOF Detector Preparation
Progress – new FADC & DAQ installed

On 03/16 (Thu), 
2 more FADCs and TCB are arrived.

NKFADC – 500MHz data sampling
2V/12bit dynamic range
4ch/mod * 2mod

TCB – ch1~20 (FADC)

Also, new DAQ system which is used in IBS 
is installed.
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TOF Detector Preparation
3bar2trig test with NKFADC

From 03/17(Fri) to 03/22(Web), 
About 1,500 cosmic ray signals are selected and saved which have 8ch coincidence.

2 trig are situated at center(0cm).

~ 12 counts/hour
~ 300 counts/day

6ch : attached to 3bar(H7195)
2ch : attached to 2trig
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TOF Detector Preparation
Reflector test, with kyung-ro

2 plastic scintillation bars are shielded
by (aluminum foil) + (thin black sheet).

After preparation of BNC, HV cables,
Test will be proceed without triggers.

DAQ will be done by what we’ve already 
had – FADC500-ibs, fdaq.

~ 0.02mm ~ 0.1mm

~ 0.2mm

~ 0.1mm
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TOF Detector Preparation
CERN Delivery preparation

Got documents from Yves yesterday.

He and Patrice think that early July is
good for installation of TOF in CERN.

To do,
1. Fill documents and send them to CERN
2. Order halogen-free cable & connector

(SMA? BNC?)
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Server status 
Backup problem

HEP2009 is now very unstable – it is disconnected(Input/Output error) by “ls” or “ll” command.
Trying to do backup again, but…
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